Author's response to reviews

Title: Macronutrients, vitamins and minerals intake and risk of esophageal squamous cell carcinoma: a case-control study in Iran

Authors:

Mahsa Jessri (jessri@ualberta.ca)
Bahram Rashidkhani (b_rashidkhani@sbmu.ac.ir)
Bahareh Hajizadeh (bhhajizadeh@yahoo.com)
Maryam Jessri (m.jessri@uq.edu.au)
Carolyn Gotay (carolyn.gotay@ubc.ca)

Version: 7 Date: 16 December 2011

Author's response to reviews: see over
1. Major revisions (we require the author to make these changes)
   Authors' contributions: Please indicate all authors have read and approved the final manuscript.
   Corrected

2. Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
   Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   Corrected